Author's response to reviews

Title: Evaluation of Anthelmintic potential of the Ethiopian Medicinal Plant Embelia schimperi Vatke in vivo and in vitro against some Intestinal Parasites

Authors:

Yared Debebe (yare9999@yahoo.com)
Mesfin Tefera (mesfin_07@yahoo.com)
Walelign Mekonnen (WalelignM@yahoo.com)
Dawit Abebe (dawisoul@gmail.com)
Samuel Woldekidan (Samuelwoldekidan@rocketmail.com)
Abiy Abebe (abiyabq@yahoo.com)
Temesgen Menberu (temeadonay@gmail.com)
Yehualashet Belete (yehualashete.belete3@gmail.com)
Bethelhem Belayneh (betelhemb7@gmail.com)
Berhanu Tesfaye (rberhanu@yahoo.com)
Ibrahim Nasir (ibranaser@yahoo.com)
Kidist Yirsaw (Y.kidist@yahoo.com)
Hirut Basha (hirut_basha@yahoo.com)
Asrat Dawit (dawitasrat53@yahoo.com)
Asfaw Debella (asfawdebella@gmail.com)

Version: 4
Date: 29 May 2015

Author's response to reviews: see over
Dear Editor

We the authors of the manuscript entitled “Evaluation of Anthelmintic potential of the Ethiopian Medicinal Plant Embelia schimperi Vatke in vivo and in vitro against some Intestinal Parasites” with MS ID: 2079151901541793 have carefully reviewed and accommodated the comments suggested by the reviewer. We have given responses to the issues addressed by the reviewers line by line.

We are ready to accommodate any further comments which might arise from the revised manuscript.

Warmest regards,

Yared Debebe
Reviewer: Lyndy McGaw

Minor essential revisions:

1. The authors mention that they have previously published the method for isolation of embelin but have not referred to this in the Methods section.

Response: The previously published method is now properly cited in the method section and the entire referencing are now corrected following the amendments made.

2. Also in the methods section, second paragraph of the sub-section “In vitro Anthelmintic activity”, change “raid” to “rapid”.

Response: Change has been made according to the comment.

3. The manuscript would still benefit from revision by a native English speaker

Response: The manuscript is now edited by native English speaker who is a UK citizen.